Author’s response to reviews

Title: Reliability of length measurements collected by community nurses and health volunteers in rural growth monitoring and promotion services

Authors:

Matilda Laar (matilda.laar@mail.mcgill.ca)

Grace Marquis (grace.marquis@mcgill.ca)

Anna Lartey (aalartey@hotmail.com)

Katherine Gray-Donald (katherine.gray-donald@mcgill.ca)

Version: 2 Date: 04 Dec 2017

Author’s response to reviews:

Dear Editor,

We thank you for the opportunity to resubmit our manuscript. We have made the recommended formatting changes and edits to the "Declarations" section.

It is our hope that the revisions meet your standards and that the paper would be published in the BMC Health Services Research. We look forward to working with you towards a final published product.

Sincerely,

Matilda Essandoh Laar, PhD